Author's response to reviews

Title: Memory and Executive Screening (MES): a brief cognitive test for detecting mild cognitive impairment

Authors:
  qihao guo (dr.guqihao@126.com)
  bin zhou (zhoubin@tri-kobe.org)
  qianhua zhao (applenasa@gmail.com)
  bei wang (bblbbbl@163.com)
  zhen hong (profzhong@sina.com)

Version: 4 Date: 10 July 2012

Author's response to reviews: see over
Jul 10, 2012

Deesha Majithia
Executive Editor
BMC Neurology

Dear Professor Deesha Majithia

On behalf of my coauthors and myself, I am hereby submitting the enclosed revised manuscript entitled “Memory and Executive Screening (MES): a brief cognitive test for detecting mild cognitive impairment” (4789663516938848) for your consideration for publication in the BMC Neurology. Please note the language had been polished according to the suggestion with the help of the editing office that suggested by BMC.

We believe that this paper will be of great interest to the readers of BMC Neurology. Dementia, especially Alzheimer’s disease is a leading cause of death in developing countries as well as developed countries. If Alzheimer’s dementia can be identified and receive proper medications at its early stage, the prognosis of these patients will be better. In this paper, we investigate the applicability of the EMS in Chinese elderly patients with amnestic mild cognitive impairment (aMCI) and Alzheimer’s dementia (AD). We found that this version is applicable for the Chinese population and it has good discrimination for MCI and AD.

Please note that this work is not currently under review by another journal and has not been previously published elsewhere either in printed or electronic form. All authors have participated in this research and have reviewed and approved this manuscript.

All correspondence regarding the manuscript may be addressed to me. And at the
same time please email a copy to Qi-hao Guo (email: dr.guoqihao@126.com)

We look forward to your consideration and review of our manuscript.

Sincerely yours,

Zhen Hong

Director
Institute of Neurology, Fudan University
#12 Wulumuqi Zhong Rd.
Shanghai 200040, P.R.China
Tel. & Fax: 86-21-62481930
Email: profzhong@sina.com